Author's response to reviews

Title: A prospective registry of emergency department patients admitted with infection.

Authors:

    Julian Williams (julian_williams@health.qld.gov.au)
    Jaimi H Greenslade (jaimi_greenslade@health.qld.gov.au)
    Juliet McKenzie (Juliet_McKenzie@health.qld.gov.au)
    Kevin Chu (kevin_chu@health.qld.gov.au)
    Anthony Brown (Afi.brown@uq.edu.au)
    David Paterson (David.antibiotics@gmail.com)
    Jeffrey Lipman (j.lipman@uq.edu.au)

Version: 3 Date: 17 January 2011

Author's response to reviews:

To the BMC Series editorial production time

Please find attached the latest version of the manuscript (ID 2769864564957105). As requested we have made the following changes.

1) The author list now matches the list in the submission system
2) The author contribution section now states that all authors have read and approved the final manuscript.
3) The references are now cited in consecutive numbers with the punctuation after the reference.

Minor revisions
1) The title is now in sentence case
2) the text "Study protocol" has now been removed
3) The affiliations section now includes the city
4) Reference 20 is formatted in accordance with the in press reference on the BMC website.
5) Vertical lines have been removed from the tables
6) Shading has now been removed from the tables.

Kind Regards

Julian Williams and Jaimi Greenslade